
• Wednesday: DVM
• Thursday : Vladimir

Friday:  Keith•
Saturday:  Steve•
Sunday: Jim•
Monday/Tuesday:  Keith•

On-call

○ Overnight, after $21 was out of the TLG, Stacked 27.5mA/hr with a production of 23.5 with 
7.8 tp on target

Stacked 26.6mA/hr with a produciton of 23.6 pbars/Mp with 7.78 Tp on target•

○ Transfer 2125 had a 4.39 horizontal TBT oscillation.  Emittances were small, so transfer 
efficiency ok.

• Unstacked 325E10 in 39 transfers over 13 sets with an average overall efficiency of 95.5%

Stacking and Tranfers

• Stacking and transfers  came back to pre-shutdown levels with relative ease.
• 10am established stacking and got one iteration so stacking and transfers before the F-Sector 

access.
• DRF1-7 had a driver power supply failure.  Just prior to this it was noticed that the station was out 

of tune, which was likely an early indication of the driver power supply going bad.
• There were several trips of the P2 permit.  M:HV102 looks like it went for a walk, but corrected 

itself.  We'll keep an eye on it.

○ 15:13
17:47○

• Pbar sequencer having problems ACL script stalling on "G:TLGSEQ is off or too close to end of the 
supercycle, waiting on $00"

20:58○

• D:Q926 DC overcurrent trip, reset and on.

○ 21:37
• MI TBT oscillation 4.39mm horizontal on pbar transfer

D:IKIK arcing PFV•

 Original Monitor (http://ap0sightglass-2.fnal.gov/view/viewer_index.shtml?id=8233

New Monitor (http://131.225.138.84/view/viewer_index.shtml?id=21)

We now have a second camera to monitor ○

○ The camera images show that the reserviour tank level from yesterday is unchanged, 
indicating that we are not leaking!

○ Auto-venting is also another sign that we are not leaking!
○ CW isolation valve script appears to be working.   We will be moving this script to a more 

permanent location today.

Dump Water System•

• D:DPWP3

Interesting Operational Happenings

○ Pbars stacked: 390.69 E10
• Stacking

Numbers
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○

○ Time stacking: 15.52 Hr
Average stacking rate: 25.17 E10/Hr○

○ Number of pulses while in stacking mode: 22314
○ Number of pulses with beam: 21245

○ Fraction of up pulses was: 95.21%

• Uptime

○ Corrected time stacking: 14.78 Hr
○ Possible average stacking rate: 26.44 E10/Hr
○ Could have stacked: 410.35 E10/Hr

• The uptime's effect on the stacking numbers

Pbars sent to the Recycler: 358.50 E10○

○ Number of transfers : 42
○ Number of transfer sets: 15
○ Average Number of transfer per set: 2.80

○ Time taken to shoot including reverse proton tuneup: 00.15 Hr
○ Transfer efficiency: 94.61%

• Recycler Transfers

○ Average POT : 7.74 E12
○ Average production: 23.76 pbars/E6 protons

• Other Info

Plots
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18:00 until this morning ($29's only)
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Since turning on 

Shutdown is at transfer 23786
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